

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (<http://bmjopen.bmj.com/site/about/resources/checklist.pdf>) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Securing All intraVenous devices Effectively in hospitalised patients - The SAVE Trial: study protocol for a multi-centre, randomised controlled trial.
AUTHORS	Rickard, Claire; Marsh, Nicole; Webster, Joan; Playford, E; McGrail, Matthew; Larsen, Emily; Keogh, Samantha; McMillan, David; Whitty, Jennifer; Choudhury, Md; Dunster, Kimble; Reynolds, Heather; Marshall, Andrea; Crilly, Julia; Young, Jeanine; Thom, Ogilvie; Gowardman, John; Corley, Amanda; Fraser, John

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.